Author’s response to reviews

Title: ADEQUACY OF CONTROL OF CARDIOVASCULAR RISK FACTORS IN AMBULATORY PATIENTS WITH TYPE 2 DIABETES ATTENDING DIABETES OUTPATIENT CLINIC AT A COUNTY HOSPITAL, KENYA

Authors:

Frederick C F Otieno (cfotieno@gmail.com)
Elijah Ogola (onogola@gmail.com)
Mercy Kimando (mkimando@gmail.com)
Kenn Mutai (mutaikk@gmail.com)

Version: 5 Date: 14 Oct 2017

Author’s response to reviews:

Dear Editorial Manager,

I have addressed the "Abstract", removed "Objective".

The 'Introduction' has been changed to 'Background'

A paragraph of "conclusion" has been added at the end of the 'Discussion'.

The Tables 1,2,3,4 and Figure 1 have been presented as separate files for each.

Declaration has been re-organised as required.

Authorship role of ENO has been given rightful description.
The raw data was submitted much earlier, however we are now using it to write a manuscript on Kidney aspects of the study. It can still be made available on request.

Signed, FCF Otieno.